Author’s response to reviews

Title: BMI and dental caries in children and adolescents: a systematic review of the literature

Authors:

Merrilyn Hooley (merrilyn@deakin.edu.au)
Helen Skouteris (helen.skouteris@deakin.edu.au)
Cecile Boganin (cecile.boganin@deakin.edu.au)
Julie Satur (juliegs@unimelb.edu.au)
Nicky Kilpatrick (nicky@bassdata.com.au)

Version: 6 Date: 22 August 2012

Author’s response to reviews: see over
August 22nd, 2012

Dr Devan Kansagara,
Associate Editor
Systematic Reviews
c/o BioMed Central
236 Gray's Inn Road
London WC1X 8HB
United Kingdom

Re: MS: 2134866555663594 BMI and dental caries in children and adolescents: a systematic review of the literature

Dear Dr Kansagara,

We thank you again for your thoughtful feedback on our earlier draft and appreciate the opportunity to resubmit for your consideration. We have followed your suggestions and concerns and submit the final manuscript for your consideration.

Kind regards,
Merrilyn Hooley